Author’s response to reviews

Title: Implementation Study of a 5-component Pediatric Early Warning System (PEWS) in an Emergency Department in British Columbia, Canada, to inform provincial scale up

Authors:

Theresa McElroy (theresa.mcelroy@cw.bc.ca)
Erik Swartz (erik.swartz@vch.ca)
Kasra Hassani (kasra.hassani@cw.bc.ca)
Sina Waibel (sina.waibel@cw.bc.ca)
Yasmin Tuff (yasmintuff@gmail.com)
Catherine Marshall (catherine.marshall@vch.ca)
Richard Chan (richard.chan@vch.ca)
David Wensley (dwensley@cw.bc.ca)
Maureen O’Donnell (modonnell@cw.bc.ca)

Version: 2 Date: 01 Nov 2019

Author’s response to reviews:

Dear editor,

Thank you for your feedback on our manuscript. We looked back at our funding statement and compared it with similar work done by other researchers. Realizing that our study did not receive any specific funding (all authors conducted the work as part of their regular jobs), we changed the funding statement to reflect that by saying: The authors received no specific funding for this work. We hope that this responds to your request.

Sincerely,
Theresa McElroy